Author’s response to reviews

Title: How Are Medical Groups Identified as High-Performing? The Effect of Different Approaches to Classification of Performance

Authors:
Sangeeta Ahluwalia (sahluwal@rand.org)
Cheryl Damberg (damberg@rand.org)
Ann Haas (ahaas@rand.org)
Paul Shekelle (shekelle@rand.org)

Version: 2 Date: 18 Jun 2019

Author’s response to reviews:

We have moved the List of Abbreviation to after the Conclusion, before the Declaration section. We have added a line in our limitations section to the possibility of chance as suggested by the peer reviewer, and as requested by the editor, provide a clean copy of the manuscript with this submission (removing the revision letter, also as requested).